Author’s response to reviews

Title: What do bereaved parents want from professionals after the sudden death of their child? A structured narrative review of the literature.

Authors:

Joanna J Garstang (joanna.j.garstang@warwick.ac.uk)
Frances E Griffiths (F.E.griffiths@warwick.ac.uk)
Peter Sidebotham (P.sidebotham@warwick.ac.uk)

Version: 2
Date: 15 May 2014

Author’s response to reviews:

The Editor
BMC Pediatrics
15 May 2014

Dear Sir,

What do bereaved parents want from professionals after the sudden death of their child? A structured narrative review of the literature.

I have made the following changes to the manuscript:

1. I have commented on ethical approval in the methods section at line 81, as this is a literature review ethical approval was not needed.
2. Line numbering has been used
3. The competing interests section has been completed fully
4. Tables 2, 3 and 4 have been included in the manuscript. Tables 5 and 6 remain as additional files
5. The manuscript has been formatted as requested.

Yours faithfully

Dr Joanna Garstang
NIHR Doctoral Research Fellow
Division of Mental Health and Wellbeing
University of Warwick